Author’s response to reviews

Title: Knowledge translation tool to improve pregnant women’s knowledge awareness of gestational weight gain goals and the risks of gaining outside recommendations: a non-randomized intervention study

Authors:

Sarah D McDonald (mcdonals@mcmaster.ca)
Christina K Park (parkku2@mcmaster.ca)
Eleanor Pullenayegum (eleanor.pullenayegum@sickkids.ca)
Kenya Bracken (bracken@mcmaster.ca)
Wendy Sword (sword@mcmaster.ca)
Helen McDonald (mcdonal@mcmaster.ca)
Binod Neupane (neupanbp@mcmaster.ca)
Valerie H Taylor (valerie.taylor@wchospital.ca)
Joseph Beyene (beyene@mcmaster.ca)
Valerie Mueller (muellevm@mcmaster.ca)
Melissa Brouwers (mbrouwer@mcmaster.ca)

Version: 4 Date: 7 April 2015

Author’s response to reviews: see over
April 6, 2015

Dr. Peter O'Donovan
Executive Editor
BMC Pregnancy and Childbirth

Dear Dr. O'Donovan,

We are pleased to submit a revised version of our paper, Manuscript ID 1869116072143551 titled "Knowledge translation tool to improve pregnant women’s awareness of gestational weight gain goals and risks of gaining outside recommendations: a non-randomized intervention study." We appreciate the constructive comment from you. Please see our response below.

Yours Sincerely,

Sarah D. McDonald, MD, MSc
Departments of Obstetrics and Gynecology (Division of Maternal-Fetal Medicine), Clinical Epidemiology and Biostatistics, and Radiology
McMaster University, HSC 3N52B
1280 Main St. West
Hamilton, Ontario, Canada L8S 4K1
Tel: 905-525 9140 ext 26559, Fax: 905-524-2911
mcdonals@mcmaster.ca

Editorial Comment

Please clarify in the manuscript whether written informed consent was obtained from participants.

Response: Thank you for this comment. All participants provided written informed consent prior to inclusion into the study, and this detail was added to the manuscript in lines 158-59.